Author’s response to reviews

**Title:** Evaluating the effectiveness of IV iron dosing for anemia management in common clinical practice: Results from the Dialysis Outcomes and Practice Patterns Study (DOPPS)

**Authors:**

Bruce Robinson (bruce.robinson@arborresearch.org)

Maria Larkina (maria.larkina@arborresearch.org)

Brian Bieber (brian.bieber@arborresearch.org)

Werner Kleophas (Werner.Kleophas@davita-dialyse.de)

Yun Li (yunlisph@umich.edu)

Francesco Locatelli (f.locatelli@ospedale.lecco.it)

Keith McCullough (keith.mccullough@arborresearch.org)

Jackie Nolen (jackiegnolen@gmail.com)

Friedrich Port (friedrich.port@arborresearch.org)

Ronald Pisoni (ronald.pisoni@arborresearch.org)

Jennifer McCready-Maynes (jennifer.mccready@arborresearch.org)

**Version:** 2  **Date:** 04 Oct 2017

**Author’s response to reviews:**

We have made the minor formatting changes requested. Regarding point #3 (Dr. Kleophas' authorship role), we have confirmed with him that it was an oversight on his part, and that he did participate in all areas needed to be considered an author. We have updated this information in the paper.